Reviewer #1:
The paper deals with an interesting issue. The manuscript is well structured and all sections are extensively organized.

Response: Thanks.

Dear Editor
The World Journal of Transplantation

Date 18/10/2023

World Journal of Transplantation Manuscript Revision-Manuscript NO- 87040

Dear Editor,

I am very grateful for your kind consideration and acceptance of our work. Please find attached the revised documents as per your recommendations as follows:

- The numbering of the figures changed as per the recommendations
- We confirm the figures are original (i.e. generated de novo by the author(s) for this paper.
- The table changed as per the recommendations
- Fingers uploaded as PPT
- Title changes

Please let us know if you need any further information.

Kind regards,

Abbas Ghazanfar